

Citation: Rahimi K, Mohseni H, Otto CM, Conrad N, Tran J, Nazarzadeh M, et al. (2017) Elevated blood pressure and risk of mitral regurgitation: A longitudinal cohort study of 5.5 million United Kingdom adults. PLoS Med 14(10): e1002404. https://doi.org/10.1371/journal.pmed.1002404

Academic Editor: Aziz Sheikh, Edinburgh University, UNITED KINGDOM

Received: June 3, 2017

Accepted: September 8, 2017

Published: October 17, 2017

Copyright: © 2017 Rahimi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data can be obtained directly from CPRD subject to the custodian's policies for scientific, data governance, and financial approvals (see www.cprd.com).

Funding: The study was funded by grants from the Oxford Martin School, the National Institute for Health Research, and the British Heart Foundation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

RESEARCH ARTICLE

Elevated blood pressure and risk of mitral regurgitation: A longitudinal cohort study of 5.5 million United Kingdom adults

Kazem Rahimi^{1,2}*, Hamid Mohseni¹, Catherine M. Otto³, Nathalie Conrad¹, Jenny Tran¹, Milad Nazarzadeh^{4,5}, Mark Woodward^{1,6,7}, Terence Dwyer¹, Stephen MacMahon^{1,6}

1 The George Institute for Global Health, University of Oxford, Oxford, United Kingdom, 2 Oxford University Hospitals NHS Foundation Trust, Oxford, United Kingdom, 3 University of Washington, Seattle, Washington, United States of America, 4 The Collaboration Center of Meta-analysis Research, Sabzevar University of Medical Sciences, Sabzevar, Iran, 5 Torbat Heydariyeh University of Medical Sciences, Torbat Heydariyeh, Iran, 6 The George Institute for Global Health, University of Sydney, Sydney, Australia, 7 Department of Epidemiology, Johns Hopkins University, Baltimore, Maryland, United States of America

* kazem.rahimi@georgeinstitute.ox.ac.uk

Abstract

Background

Mitral regurgitation in people without prior cardiac disease is considered a degenerative disease with no established risk factors for its prevention. We aimed to test the hypothesis that elevated systolic blood pressure (SBP) across its usual spectrum is associated with higher risk of mitral regurgitation.

Methods and findings

We used linked electronic health records from the United Kingdom Clinical Practice Research Datalink (CPRD) from 1 January 1990 to 31 December 2015. CPRD covers approximately 7% of the current UK population and is broadly representative of the population by age, sex, and ethnicity. About 5.5 million UK patients with no known cardiovascular or valve disease at baseline were included in this cohort study. We investigated the relationship between blood pressure (BP) and risk of mitral regurgitation using Cox regression models. Our primary exposure variable was SBP and our primary outcome was incident reports of mitral regurgitation, which were identified from hospital discharge reports or primary care records.

Of the 5,553,984 patients in the CPRD that met our inclusion criteria, during the 10-year follow-up period, 28,655 (0.52%) were diagnosed with mitral regurgitation and a further 1,262 (0.02%) were diagnosed with mitral stenosis. SBP was continuously related to the risk of mitral regurgitation with no evidence of a nadir down to 115 mmHg (p < 0.001). Each 20 mmHg increment in SBP was associated with a 26% higher risk of mitral regurgitation (hazard ratio [HR] 1.26; CI 1.23, 1.29). The observed association was partially mediated by diseases affecting the left ventricle during follow-up (myocardial infarction [MI], ischaemic heart disease [IHD], cardiomyopathy, and heart failure). However, the percentage of excess risk mediated (PERM) by these proximate causes of secondary mitral regurgitation was only

Competing interests: KR receives a stipend as a specialty consulting editor for PLOS Medicine and serves on the journal's editorial board. MW is a consultant to Amgen on analyses of routinely collected US health data. JT has received funding from Rhodes scholarship for DPhil study at The University of Oxford. The Rhodes scholarship has no role in the study design or related decisions; JT is director of CHASE (non-government not-forproft organisation) and previous Director of Libromat (non-government not-for-proft organisation) and secretariat of the Academy of Medical Sciences working group for multimorbidity. NC is supported by a grant from the British Heart Foundation. All the other authors have declared that no competing interests exist.

Abbreviations: BMI, body mass index; BP, blood pressure; CPRD, Clinical Practice Research Datalink; DBP, diastolic blood pressure; HDL, highdensity lipoprotein; HR, hazard ratio; IHD, ischaemic heart disease; IQI, interquartile interval; LDL, low-density lipoprotein; MI, myocardial infarction; PERM, percentage of excess risk mediated; PP, pulse pressure; SBP, systolic blood pressure. 13% (CI 6.1%, 20%), and accounting for them had little effect on the long-term association between SBP and mitral regurgitation (mediator-adjusted HR 1.22; CI 1.20, 1.25; p < 0.001). Associations were similar for each 10 mmHg increment in diastolic blood pressure (DBP) (p < 0.001) or each 15 mmHg increment in pulse pressure (PP) (p < 0.001). By contrast, there was no association between SBP and risk of mitral stenosis (HR per 20 mmHg higher SBP 1.03; CI 0.93, 1.14; p = 0.58). These analyses are based on routinely collected data from health records which may be sensitive to measurement errors, and the observed associations may not be generalizable to less severe and subclinical cases of mitral regurgitation.

Conclusions

Long-term exposure to elevated BP across its whole spectrum is associated with an increased risk of primary and secondary mitral regurgitation. These findings suggest that BP control may be of importance in the prevention of mitral regurgitation.

Author summary

Why was this study done?

- Elevated blood pressure (BP) is a major risk factor for a range of cardiovascular conditions.
- Mitral regurgitation in people without prior cardiac disease is considered a degenerative disease with no established risk factors for its prevention.
- Whether elevated BP is also a risk factor for mitral regurgitation is unknown.

What did the researchers do and find?

- About 5.5 million UK patients with no known cardiovascular or valve disease at baseline were included in this cohort study and were followed up for about 10 years.
- This study shows a clear log-linear association between elevated BP and risk of mitral regurgitation (but not mitral stenosis) with no apparent threshold below or above which the relationship ceased to exist.
- The association was only partially mediated by conditions that are established causes of secondary mitral regurgitation, suggesting that elevated BP has a direct and independent effect on valve degeneration.

What do these findings mean?

- The findings from this study suggest that mitral regurgitation is not an inevitable consequence of ageing and that its risk might be modifiable with established interventions.
- Future studies should establish the causality of the observed associations and test whether BP lowering might reduce the risk of mitral regurgitation.

Introduction

Mitral regurgitation is the most common valvular heart disorder in high-income countries, and its prevalence increases with age [1,2]. The strong age-dependent prevalence rates suggest that, as a consequence of worldwide ageing and population growth, crude rates of mitral regurgitation are likely to increase further in the coming decades.

Despite substantial progress in our understanding of the pathophysiology of mitral regurgitation and advances in surgical and interventional valve replacement therapies, there are no established preventive strategies [3]. Causes of mitral regurgitation are categorised as "primary" when due to abnormalities of the valve leaflets and chordae or as "secondary" when due to distortion of ventricular shape related to ischaemic heart disease (IHD) or a cardiomyopathy. However, the distinction between these subtypes is not always obvious, and about twothirds of all mitral regurgitation cases are classified as degenerative [4], implying that they are a natural consequence of ageing with no possibility of altering their course.

Elevated blood pressure (BP) is a strong risk factor for a range of cardiovascular conditions [5]. Given that increased BP correlates with higher left ventricular pressure, and this, in turn, exposes the mitral valve to higher physical stress, it seems plausible that long-term exposure to higher BP could also lead to structural and functional changes of the mitral valve. A cross-sectional analysis of the Framingham study showed a positive association between hypertension and mitral regurgitation [6]. However, to our knowledge, there are no longitudinal studies that have reported associations. We, therefore, aimed to test the hypothesis that elevated systolic blood pressure (SBP) across its usual spectrum is associated with higher risk of mitral regurgitation.

Methods

Data source

We used linked electronic health records from the UK Clinical Practice Research Datalink (CPRD) from 1 January 1990 to 31 December 2015. The CPRD contains anonymised patient data from 674 general practices in the UK [7]. It covers approximately 7% of the current UK population and is broadly representative of the population by age, sex, and ethnicity. It partially links primary care records with discharge diagnosis from secondary care (Hospital Episode Statistics) and mortality data from national death registries (Office for National Statistics). The dataset is considered the most comprehensive longitudinal primary care database with serial collection of information relating to diagnosis, treatments, investigations, and outcomes [8] and has been validated for epidemiological research for a range of conditions, including those heavily relying on imaging tests [7,9]. Scientific approval for this study was given by the CPRD Independent Scientific Advisory Committee (ISAC). The CPRD Group has obtained ethics approval from a National Research Ethics Service Committee for all purely observational research using anonymised CPRD data. A separate ethics approval was not required for this work.

Study population

A total of 6,613,644 patients between 30 and 90 years old with at least 1 BP measurement were identified. Patients entered the study at the date of their earliest BP measurement (baseline) and exited the study at the earliest date of transfer out of the general practice, death, end of the study period, or a record of mitral regurgitation. We further excluded all individuals who at baseline had a prior clinical diagnosis of cardiovascular disease (409,075), mitral valve disease including mitral valve prolapse (14,592), or were prescribed lipid-lowering or antihypertensive

medication (196,451). Cardiovascular disease was defined, as previously reported [10,11], using ICD-10 and Read codes for myocardial infarction (MI), IHD (including reports of angina or coronary revascularization), stroke, transient ischaemic attack, heart failure, cardiomyopathy, chronic kidney disease, peripheral arterial disease, atrial fibrillation, or venous thromboembolism. We further excluded individuals with extreme values of baseline SBP with recordings outside the range 50 to 300 mmHg (14,233; 0.08% of all recordings, but their inclusion led to virtually unchanged associations). Patients with less than 1 year follow-up with the general practice (126,358) and patients whose BP measurements were before 1990 (313,633) were also excluded. This led to the selection of 5,553,984 individuals for this study with a median (interquartile interval [IQI]) follow-up duration of 10 (4.7, 17) years.

Outcomes and exposures

Our primary outcome was incident reports of mitral regurgitation, which were identified from hospital discharge reports, death registers, or primary care records using the diagnostic codes shown in S1 Table. We excluded diagnostic codes which classified mitral valve disease as congenital or when mixed mitral valve disease was reported with no clear indication of the dominant condition. We did not have complete information on echocardiography for all patients. However, previous validation studies based on electronic health records have shown that the majority of clinically recorded valve disease codes are based on echocardiographic assessments and have a high positive predictive value for the valve diagnosis (>85%) [2]. They further show that recorded cases typically represent moderate to severe severity ratings that are clinically relevant [2] (whereas epidemiological studies with population screening tend to identify a larger number of patients with mild and asymptomatic diseases [12]). For more direct validation of valve disease and its severity in our study, we performed 3 subgroup analyses (see Statistical analysis). In addition, as a negative control secondary outcome, we chose incident reports of mitral stenosis, expecting no association with BP, since mitral stenosis without regurgitation is nearly always due to rheumatic valve disease. Finally, to test the validity of database and methods for measuring and modelling exposure variables and covariates, we chose incident stroke as a positive control outcome and report BP associations with stroke.

BP measurements were identified from the CPRD database as measurements with an entity type 1 (the entity type corresponding to a BP measurement). We defined SBP as our primary exposure because SBP has been shown to have the strongest predictive ability among most other measures of BP for most cardiovascular outcomes [13,14] and because it is a better indicator of afterload than diastolic blood pressure (DBP) and hence could play a stronger role in the pathophysiology of mitral regurgitation. However, in additional prespecified analyses, we chose DBP and pulse pressure (PP; defined as SBP–DBP) as alternative exposure variables to investigate any differential effect of BP indices on risk and to enable comparison of our findings with epidemiological studies of BP associations with other outcomes [13–15].

Statistical analysis

In consideration of measurement error surrounding a single BP measurement and other timedependent fluctuations of BP (in particular in the context of routinely recorded BP by doctors and nurses), we used multiple BP measurements to calculate 'usual' values that are corrected for regression dilution [16]. As previously reported [10,11,17], we used generalised estimating equations to regress serial BP measurements within the median follow-up (mean of 6.7 measurements per patient) on the baseline BP. Usual SBP, DBP, and PP were analysed as continuous variables with 20 mmHg, 10 mmHg, and 15 mmHg increments, respectively. SBP was further analysed as a categorical variable (\leq 120 mmHg, 121 mmHg to 140 mmHg, 141 mmHg to 160 mmHg, and >160 mmHg), modelled as a continuous variable with 1 mmHg increments, and plotted using Cox regression with restricted cubic spline with 4 knots. We determined the association between SBP (and other BP indices) and each outcome using Cox regression, with hazard ratios (HRs) displayed with floating absolute risks, which do not require the selection of an arbitrary baseline group for display of confidence intervals [18]. We tested the proportional hazards assumption of Cox models using log cumulative hazards plots for SBP categories and found no evidence of lack of proportional hazards (S1 Fig). Our primary analyses were adjusted for baseline sex, age, body mass index (BMI, kg/m²), smoking status (no, previous, yes), cholesterol (total, low-density lipoprotein [LDL], high-density lipoprotein [HDL]), and year of the initial BP measurement (as a categorical variable [1990 to 1994, 1995 to 1999, 2000 to 2004, 2005 to 2009, and 2010 to 2013]) to control for potential cohort effects, and stratified by general practice. For BMI, smoking, and lipid levels, we utilised the most recent measurement within 2 years of the baseline SBP measurement (index encounter with the general practice when SBP was recorded). If no measurement was available within this timeframe, we classified the covariate as missing. We addressed missing data using multiple imputation by expectation-maximisation with bootstrapping [19], generating 5 imputations [20].

To investigate the extent to which the potential association between BP and mitral regurgitation is mediated by incident MI, IHD, heart failure, or cardiomyopathy, which are established causes of secondary mitral regurgitation, we performed time-varying adjustments for these events during the follow-up. The HR from this confounder- and mediator-adjusted model provides an estimation of association between BP and primary mitral regurgitation. To determine what proportion of the association between usual BP and risk of mitral regurgitation was mediated by these events, we then calculated the percentage of excess risk mediated (PERM) using the formula: PERM = (HR_{confounder-adjusted} – HR_{confounder- and mediator-adjusted})/ (HR_{confounder-adjusted}). We estimated confidence intervals through percentile bootstrapping based on 1,000 runs.

We performed subgroup analyses and report interactions by age, sex, and BMI groups. We also compared the predictive value of PP, mean arterial pressure, and mid arterial pressure on the risk of mitral regurgitation using Harrell's c-statistic [21]. Mean arterial pressure was defined as 2/3 DBP + 1/3 SBP and mid arterial pressure as 1/2 SBP + 1/2 DBP.

To investigate the validity of our outcome variables and the potential effect of valve severity on outcomes, we performed the following 3 sensitivity analyses. First, we extracted information on echocardiograms within 14 days prior to the recorded diagnosis of mitral regurgitation and stratified analyses by echocardiographically supported diagnoses of mitral regurgitation versus other reports with no recorded information on diagnostic tools. Second, we stratified our analyses according to whether or not there was a definitive report of valve replacement therapy (i.e., surgical or interventional valve replacement or repair) to assess whether observations might have been diluted due to inclusion of mild or asymptomatic cases. Third, we stratified our analyses by initial source of case reporting (hospital discharge versus outpatient or primary care), assuming hospital diagnosis cases to be more severe and more likely to be verified by specialists. Additional sensitivity and validation analyses are described in S1 Text.

Study findings are reported in accordance with the reporting of studies conducted using Observational Routinely-collected health Data (RECORD) Statement [22]. There was no specific analysis plan for the current study, but the main analysis followed a general analysis plan for investigation of associations between BP and several outcomes including valvular heart disease (S2 Text and earlier reports [10,11,17,23] describing the general analysis plan). Deviations from the general protocol were largely informed by the clinical differences between outcomes and reviewer comments. More specifically, for this paper, we introduced a mediation analysis





Fig 1. Adjusted age-specific HRs of SBP for mitral regurgitation. Age bands refer to age at baseline. Adjustments were for sex, BMI, smoking, year of BP measurement, total cholesterol, LDL cholesterol, and HDL cholesterol. Confidence intervals are displayed as floating absolute risks. HRs and 95% CI are displayed for 1 mmHg increments in SBP and plotted using Cox regression with restricted cubic spline with 4 knots, relative to the reference category (individuals aged 30–50 years with usual SBP 115–120 mmHg).

Abbreviations: BMI, body mass index; BP, blood pressure; HDL, high-density lipoprotein; HR, hazard ratio; LDL, low-density lipoprotein; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002404.g001

to assess the mediator-corrected associations and several sensitivity analyses to validate our findings (see above and <u>S1 Text</u>). In addition, the restricted cubic spline Cox regression (Fig 1) and the further division of age groups in Fig 3 were specific requests by reviewers of this journal. Statistical analyses were performed in R, version 3.3 (R Foundation for Statistical Computing, Vienna, Austria).

Table 1.	Baseline characteristics and outcome rate	es by	categories o	of usual SBP.

	<121 mmHg n = 1,362,861	121–140 mmHg n = 3,669,483	141–160 mmHg <i>n</i> = 486,060	>160 mmHg <i>n</i> = 35,580	Total n = 5,553,984
Mitral valve disease, <i>n</i> (%)					
Mitral regurgitation	2,621 (0.19)	18,350 (0.5)	7,047 (1.4)	637 (1.8)	28,655 (0.52)
Mitral stenosis	143 (0.01)	796 (0.025)	298 (0.061)	25 (0.07)	1,262 (0.023)
Age categories, years, n (%)					
30–50	1,264,938 (93)	2,730,650 (74)	133,965 (28)	4,295 (12)	4,133,848 (74)
51–60	65,106 (4.8)	477,074 (13)	111,379 (23)	6,872 (19)	660,431 (12)
61–70	21,510 (1.6)	280,910 (7.7)	119,808 (25)	10,195 (29)	432,423 (7.8)
71–90	11,307 (0.83)	180,849 (4.9)	120,908 (25)	14,218 (40)	327,282 (5.9)
Age, median (IQI)	32 (30, 40)	39 (33, 52)	60 (49, 70)	67 (57, 74)	39 (33, 52)
Sex, <i>n</i> (%)					
Female	969,470 (71)	1,809,216 (49)	240,430 (49)	20,899 (59)	3,040,015 (55)
BMI (kg/m ²) categories, <i>n</i> (%)					
≤25	762,328 (72)	1,376,738 (49)	110,744 (32)	7,548 (32)	2,257,358 (54)
26–30	228,736 (22)	940,840 (34)	134,845 (40)	9,113 (39)	1,313,534 (31)
31–35	53,260 (5.0)	327,748 (12)	63,301 (19)	4,467 (19)	448,776 (11)
>35	18,183 (1.7)	142,824 (5.1)	32,416 (9.5)	2,439 (10)	195,862 (4.6)
Missing, %	30	24	30	34	24
BMI, median (IQI)	23 (21, 26)	25 (23, 28)	27 (24, 31)	27 (24, 31)	25 (22, 28)
Smoking history, <i>n</i> (%)					
Never smoked	701,039 (58)	1,815,237 (57)	223,881 (57)	16,304 (58)	2,756,461 (57)
Ex-smoker	125,566 (10)	431,724 (14)	71,424 (18)	5,261 (19)	633,975 (13)
Current smoker	373,176 (31)	939,597 (29)	96,528 (25)	6,445 (23)	1,415,746 (29)
Missing, %	12	13	19	21	13
Cholesterol (mmol/L), median (IQI)					
Total	5.0 (4.3, 5.7)	5.3 (4.6, 6.1)	5.6 (4.8, 6.3)	5.6 (4.8, 6.4)	5.3 (4.6, 6.1)
Total missing, %	92	84	71	63	85
LDL	3.0 (2.4, 3.7)	3.2 (2.6, 3.9)	3.3 (2.6, 4.0)	3.3 (2.6, 4.1)	3.2 (2.6, 3.9)
LDL missing, %	96	91	83	81	93
HDL	1.4 (1.2, 1.7)	1.3 (1.1, 1.6)	1.3 (1.1, 1.6)	1.39 (1.1, 1.7)	1.3 (1.1, 1.6)
HDL missing, %	95	90	79	76	92
Intermediary conditions, n (%)	22,279 (1.6)	197,899 (5.4)	87,223 (18)	8,814 (25)	316,215 (5.7)
Follow-up (year), median (IQI)	9.2 (4.1, 16)	10.0 (4.7, 17)	12 (6.6, 17)	11 (6.7, 16)	10 (4.7, 17)

Intermediary conditions: IHD, MI, heart failure or cardiomyopathy during follow-up.

Abbreviations: BMI, body mass index; HDL, high-density lipoprotein; IHD, ischaemic heart disease; IQI, interquartile interval; LDL, low-density lipoprotein; MI, myocardial infarction; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002404.t001

Results

Of the 5,553,984 patients in the CPRD that met our inclusion criteria, 28,655 (0.52%) were diagnosed with mitral regurgitation during follow-up and a further 1,262 (0.02%) were diagnosed with mitral stenosis. Patient characteristics by usual SBP categories are shown in Table 1.

Age-specific analyses showed a continuous and approximately log-linear adjusted relationship between usual SBP and risk of mitral regurgitation throughout the SBP range with no evidence of a threshold below or above which the associations were different (Fig 1). As expected, the HRs for a given difference in usual SBP are smaller in older age groups (as indicated by the

Patients with MR	Total	1		HR	95% CI
2621	1362676	+		1.00	[0.96; 1.04]
18350	3668540	+		1.23	[1.22; 1.25]
7047	485721		+	1.44	[1.40; 1.48]
637	35552		-+	1.49	[1.38; 1.61]
28582	5465126	+		1.26	[1.23; 1.29]
	Γ		1		
	0.75	1	1.5	2	
	Patients with MR 2621 18350 7047 637 28582	Patients with MR Total 2621 1362676 18350 3668540 7047 485721 637 35552 28582 5465126 0.75	Patients with MR Total 2621 1362676 18350 3668540 7047 485721 637 35552 28582 5465126 0.75 1	Patients with MR Total 2621 1362676 18350 3668540 7047 485721 637 35552 28582 5465126 0.75 1	Patients with MR Total HR 2621 1362676 1.00 18350 3668540 1.23 7047 485721 $#$ 637 35552 $#$ 28582 5465126 $#$ 1.26 0.75 1 1.5 2

Fig 2. HRs for mitral regurgitation by categories of usual SBP. HRs and 95% CI are displayed using floating absolute risk and are corrected for regression dilution. Models are adjusted for age, sex, BMI, smoking, calendar year, total cholesterol, LDL cholesterol, and HDL cholesterol. Abbreviations: BMI, body mass index; HDL, high-density lipoprotein; HR, hazard ratio; LDL, low-density lipoprotein; MR, mitral regurgitation; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002404.g002

slope of the age-specific associations). However, the absolute risk of developing mitral regurgitation increases substantially with age, leading to a substantial net contribution of elevated SBP on risk of mitral regurgitation even in the older age groups (Fig 1).

When usual SBP was analysed as a continuous variable, each 20 mmHg increment in usual SBP was associated with a 26% higher risk of mitral regurgitation (adjusted HR 1.26; CI 1.23, 1.29) (Fig 2). Compared with the reference category of usual SBP \leq 120 mmHg, patients with usual SBP \geq 161 mmHg were 1.5 times more likely to be diagnosed with mitral regurgitation (HR 1.49; CI 1.38, 1.61) (Fig 2).

Adjusted HRs among subgroups are shown in Fig 3. HRs attenuated with increasing age (p for trend < 0.001). For patients aged less than 50 years at baseline, each 20 mmHg increment in SBP was associated with a 54% higher risk of mitral regurgitation (HR 1.54; CI 1.46, 1.63), whereas in patients aged 71–90 the risk was higher by 13% (HR 1.13; CI 1.08, 1.19) (Fig 3). Proportional differences were broadly similar between men and women. There was evidence that the associations between usual SBP and mitral regurgitation differed by baseline BMI, with weaker associations observed among those with higher BMI (p for interaction < 0.01, Fig 3, p for trend = 0.44). Associations were similar when echocardiographically supported reports were compared with reports without evidence of an echocardiogram within 14 days before the valve diagnosis, when hospital discharge reports were compared with outpatient diagnoses recorded by general practitioners, or when definitive reports of valve replacement were compared with patients with no clear reports of valve replacement (Fig 4).

Associations of PP and DBP with mitral regurgitation were broadly consistent with those for SBP: each 15 mmHg higher usual PP was associated with a 23% greater risk of mitral regurgitation (HR 1.23; CI 1.20, 1.27), and each 10 mmHg higher usual DBP was associated with a 24% greater risk of mitral regurgitation (HR 1.24; CI 1.20, 1.28). Similar to SBP, these associations differed by age group, with stronger associations observed among the younger groups (*p* for trend < 0.001 for both DBP and PP, S2 Fig). In comparative analyses, SBP showed the strongest, and DBP the weakest, overall predictive power among the BP indices investigated (S2 Table). Age-stratified analyses were broadly consistent with the overall results (S2 Table).

	Patients with MR	Total	I	HR	95% CI
Age groups					
30–40	3701	3141134		1.59	[1.45; 1.75]
41–45	1778	488122		— 1.75	[1.56; 1.95]
46–50	2349	435642		1.37	[1.25; 1.50]
51–55	3117	356289		1.30	[1.20; 1.40]
56–60	3675	294249		1.27	[1.19; 1.36]
61–65	4233	240641		1.18	[1.11; 1.25]
66–70	3855	185574		1.17	[1.10; 1.25]
71–75	2981	136966		1.15	[1.08; 1.23]
76–80	1806	101320		1.11	[1.02; 1.22]
81–90	1087	85189		1.14	[1.02; 1.27]
Test for interacti	ion: <i>p</i> < 0.01				
Sex					
Female	13213	2473201		1.26	[1.21; 1.31]
Male	15369	2991925		1.28	[1.24; 1.32]
Test for interacti	ion: <i>p</i> = 0.48				
BMI					
<26	12284	2818397	-	1.32	[1.27; 1.37]
26–30	10773	1764748	-	1.26	[1.21; 1.31]
31–35	4184	650156		1.16	[1.09; 1.23]
>35	1341	231825		1.15	[1.04; 1.27]
Test for interacti	ion: <i>p</i> < 0.01				
		I	1 1	I	
		0.75	1 1.5	2	

Fig 3. HRs for mitral regurgitation per 20 mmHg higher usual SBP by age, sex, and BMI categories. HRs and 95% CI are displayed using floating absolute risks and are corrected for regression dilution. Models are adjusted for age (per year), sex, BMI, smoking, calendar year, total cholesterol, LDL cholesterol, and HDL cholesterol. Abbreviations: BMI, body mass index; HDL, high-density lipoprotein; HR, hazard ratio; LDL, low-density lipoprotein; MR, mitral regurgitation; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002404.g003

IHD, MI, heart failure, or cardiomyopathy during follow-up were strongly associated with mitral regurgitation (HR 2.94; CI 2.86, 3.03). Adjusting for their effect provides an estimate of the association between BP and mitral regurgitation without a mediating effect of left ventricular disease as the more proximal causes of secondary mitral regurgitation. This adjustment led to a mild attenuation of associations between SBP and mitral regurgitation (from confounder-adjusted HR 1.26; CI 1.23, 1.29 to confounder- and mediator-adjusted HR 1.22; CI 1.20, 1.25).

The percentage of excess risk attributable to these intermediary outcomes was 13% (CI 6.1% to 20%), suggesting that only 13% of the overall association across all primary and secondary mitral regurgitation reports is mediated by left ventricular disease prior to a diagnosis of mitral regurgitation. The potential effect of over-adjustment assessed by a progressive adjustment of covariates, in which consecutively adjusting for each covariate had little effect on the estimated HRs per 20 mmHg higher SBP for mitral regurgitation (S3 Fig).

	Patients with MR	Total	I	HR 95% CI
Echocardiograr	n prior to diagr	nosis		
Yes	6877	5465126		1.23 [1.17; 1.29]
Unknown	21705	5465126	+-	1.27 [1.23; 1.30]
Test for interaction:	p = 0.32			
Valve replacem	ent therapy			
Yes	1700	5465126		1.23 [1.11; 1.36]
Unknown	26882	5465126	-+-	1.26 [1.23; 1.29]
Test for interaction:	p = 0.62			
First source of	diagnosis			
Out-patient	<u> </u>	5465126	-	1.25 [1.21; 1.29]
In-hospital	11581	5465126	-	1.27 [1.22; 1.32]
Test for interaction:	p = 0.58			
		0.75	1 1.5	2

Fig 4. Subgroup analyses for mitral regurgitation per 20 mmHg higher usual SBP. Models are adjusted for sex, BMI, smoking, calendar year, total cholesterol, LDL cholesterol, and HDL cholesterol. Valve replacement therapy refers to mitral valve repair or replacement. Confidence intervals are displayed as floating absolute risks. Abbreviations: BMI, body mass index; HDL, high-density lipoprotein; HR, hazard ratio; LDL, low-density lipoprotein; MR, mitral regurgitation; SBP, systolic blood pressure.

https://doi.org/10.1371/journal.pmed.1002404.g004

In our negative control analysis, there was no association between SBP and risk of mitral stenosis (HR per 20 mmHg higher SBP 1.03; CI 0.93, 1.14) (S4–S6 Figs). Also supporting our main findings, our positive control analysis with stroke as the outcome did not show any evidence of bias towards extreme: overall each 20 mmHg increment in SBP was associated with a 34% higher risk of stroke (HR 1.34; CI 1.33, 1.35) across all age groups or a 43% higher risk of stroke (HR 1.43; CI 1.40, 1.47) among the patients aged between 50 and 60 years old. Additional sensitivity analyses produced very similar results to our main analyses (S1 Text).

Discussion

In this large-scale cohort of UK adults involving approximately 30,000 new cases of mitral regurgitation with 10 years of follow-up, we show that exposure to elevated BP was continuously associated with an increased risk of mitral regurgitation with no apparent threshold at which the association ceased to exist. More specifically, every 20 mmHg increment in usual SBP was associated with a 26% higher risk of mitral regurgitation during follow-up. Associations differed by baseline features of patients, with stronger associations observed in younger and leaner patient groups.

Mitral regurgitation is a heterogeneous condition that can be caused by primary structural abnormalities of the valve apparatus (primary mitral regurgitation) or diseases of the left ventricle leading to incomplete closure of a structurally normal valve (secondary mitral regurgitation). The main diseases of the left ventricle that are commonly considered as more proximate causes of secondary mitral regurgitation are MI, IHD, heart failure, and cardiomyopathy. Because elevated BP is a known risk factor for these conditions, it is possible that the observed association between BP and mitral regurgitation is entirely or largely a reflection of such indirect effects. However, accounting for the possible intermediary effect of diseases of the left ventricle had little impact on the observed associations between elevated BP and risk of mitral regurgitation. In confounder- and mediator-adjusted models, which in essence provide an estimation of the association between BP and primary mitral regurgitation, the overall HR was only slightly attenuated and only 13% of the excess risk could be attributed to the intermediary outcomes. This indicates that 87% of the effect of BP on mitral regurgitation was independent of diseases of the left ventricle that occurred during follow-up, suggesting that BP may exert its main effect on the mitral valve directly or via mechanisms unrelated to left ventricular dilatation or dysfunction. However, our mediation analysis was relatively crude and alternative causal inference modelling approaches might be better suited to disentangle the effect of mediators and confounders during follow-up.

Our study was not designed to investigate the underlying mechanisms of the observed associations between BP and mitral regurgitation, and we can only speculate about possible pathophysiological pathways. Physiologically, the mitral valve is exposed to considerable mechanical force in systole reflecting rapid closure of the valve at the start of systole and subsequent increased pressure difference between the left ventricle and atrium throughout systole. High BP is also associated with elevated left atrial diastolic pressures, which during diastole leads to a higher mitral valve opening pressure. This might alter blood flow patterns, resulting in abnormal shear stress on the atrial surface of the valve leaflets, as well as abnormal shear stress on the ventricular side of the leaflets when the valve is closed. Therefore, it seems plausible that the mechanical stress caused by the elevated BP will lead to gradual structural changes of the valve apparatus. Future studies could explore these and alternative mechanisms further. Such studies, ideally when repeated cardiac imaging is available, might also be able to investigate whether the observed associations differ by presence or type of underlying valve pathologies, such as Barlow disease or mitral valve prolapse [24].

We are not aware of any prior longitudinal studies that have investigated the association between BP and risk of mitral regurgitation. A cross-sectional study based on the Framingham cohort involved 2,881 participants and showed that those with hypertension had a 1.6 times higher risk of concomitant mitral regurgitation (CI 1.2, 2.0) [6]. Further evidence in support for our findings comes from studies that have reported an association between elevated BP and risk of mitral annulus calcification, which is related to degenerative mitral regurgitation [25]. Our large-scale cohort study extends these prior findings and shows that exposure to elevated BP is significantly associated with an increased the risk of mitral regurgitation.

Our analyses are based on routinely collected data from linked electronic health records, which might be more prone to measurement errors than prospectively designed research datasets. However, measurement errors in our exposure variable are unlikely to be biased, given that BP measurements were often recorded several years before the occurrence of outcomes and many of them were within the conventionally 'normal' range. This is supported by the lack of association between BP and mitral stenosis in our study (although this does not exclude a significant association in certain groups of patients with mitral stenosis, such as elderly in whom mitral stenosis may be due to mitral annulus calcification). The confirmation of the association between BP and stroke in our validation analysis further supports the suitability of our database and exposure variable for investigation of valve disease [13,14]. To reduce the impact of inevitable random measurement error in our exposure variable, we used multiple BP measurements and corrected baseline values for regression dilution. We did not adjust for anti-hypertensive use during follow-up but this and any other residual uncontrolled measurement error in BP measurements would only be expected to have biased our estimates towards the null and reduced statistical power [16,26]. In addition, our regression dilution correction method for BP measurements indirectly takes account of the effect of anti-hypertensive treatment during follow-up, consistent with prior research that has shown that additional control

for anti-hypertensive use has little impact on the regression dilution ratios [27]. As for our outcome variable mitral regurgitation, it is well known that studies based on clinically reported events are more likely to only capture functionally relevant and symptomatic disease states and, for this reason and because of incomplete record linkage, will underestimate rates compared to epidemiological studies that screen the entire population for subclinical and clinical disease. Thus, the observed associations may not be generalizable to less severe and subclinical cases of mitral regurgitation. Within the setting of clinical mitral regurgitation, any random measurement error in diagnostic coding or missingness as a result of incomplete case ascertainment would be expected to make estimates less precise and is unlikely to have any material impact on our overall findings [16]. In addition, our findings were robust to several alternative assumptions, including a comparison of in-patient and out-patient diagnoses and cases with a definitive report of echocardiogram versus others, suggesting that residual confounding is unlikely to have played a substantial role. Nonetheless, as with most observational studies, our study in isolation may not establish causation beyond doubt. New trials with mitral valve regurgitation as an outcome are needed to answer this question reliably. Alternatively or in addition, individual patient data meta-analysis of unreported valvular outcomes from previously published trials or Mendelian randomisation studies might provide additional insights into the potential causal role of BP on mitral regurgitation.

In conclusion, our large-scale prospective study suggests that long-term exposure to elevated BP across its whole spectrum is associated with an increased risk of primary and secondary mitral regurgitation. These findings suggest that BP control may be of importance in the prevention of mitral regurgitation.

Supporting information

S1 Text. Additional sensitivity analyses. (DOCX)

S2 Text. General study protocol. (DOC)

S1 Fig. Log cumulative hazards plots for SBP relating to mitral regurgitation. Abbreviation: SBP, systolic blood pressure. (DOCX)

S2 Fig. HRs for mitral regurgitation per 10 mmHg higher usual DBP or per 15 mmHg higher usual PP, by age categories. Abbreviations: DBP, diastolic blood pressure; HR, hazard ratio; PP, pulse pressure. (DOCX)

S3 Fig. HRs per 20 mmHg higher SBP for mitral regurgitation, with progressive adjustment for age, sex, calendar year, BMI, smoking, and baseline LDL cholesterol, HDL cholesteral, and total cholesterol, and stratified by practice level. Abbreviations: BMI, body mass index; HDL, high-density lipoprotein; HR, hazard ratio; LDL, low-density lipoprotein; SBP, systolic blood pressure. (DOCX)

S4 Fig. Adjusted age-specific HRs of SBP for mitral stenosis. Abbreviations: HR, hazard ratio; SBP, systolic blood pressure. (DOCX)

S5 Fig. HRs for mitral stenosis by categories of usual SBP. Abbreviations: HR, hazard ratio; SBP, systolic blood pressure.

(DOCX)

S6 Fig. HRs per 20 mmHg higher SBP for mitral stenosis, with progressive adjustment for age, sex, calendar year, BMI, smoking, and baseline LDL cholesterol, HDL cholesterol, and total cholesterol, and stratified by practice level. Abbreviations: BMI, body mass index; HDL, high-density lipoprotein; HR, hazard ratio; LDL, low-density lipoprotein; SBP, systolic blood pressure.

(DOCX)

S1 Table. Read and ICD-10 codes for mitral valve disease. (DOCX)

S2 Table. Relative ability of different BP indices to independently predict mitral regurgitation, by age categories and overall. Abbreviation: BP, blood pressure. (DOCX)

S1 RECORD Checklist. RECORD checklist. (DOCX)

Author Contributions

Conceptualization: Kazem Rahimi.

Formal analysis: Hamid Mohseni.

Funding acquisition: Kazem Rahimi.

Investigation: Jenny Tran.

Methodology: Kazem Rahimi, Hamid Mohseni, Nathalie Conrad, Jenny Tran, Milad Nazarzadeh, Mark Woodward, Terence Dwyer, Stephen MacMahon.

Project administration: Kazem Rahimi.

Resources: Kazem Rahimi, Nathalie Conrad.

Supervision: Catherine M. Otto, Mark Woodward.

Validation: Catherine M. Otto.

Writing - original draft: Kazem Rahimi.

Writing – review & editing: Hamid Mohseni, Catherine M. Otto, Nathalie Conrad, Jenny Tran, Milad Nazarzadeh, Mark Woodward, Terence Dwyer, Stephen MacMahon.

References

- 1. Coffey S, Cairns BJ, lung B. The modern epidemiology of heart valve disease. Heart. 2016; 102: 75–85. https://doi.org/10.1136/heartjnl-2014-307020 PMID: 26541169
- Andell P, Li X, Martinsson A, Andersson C, Stagmo M, Zöller B, et al. Epidemiology of valvular heart disease in a Swedish nationwide hospital-based register study. Heart. 2017; 0: heartjnl-2016-310894. https://doi.org/10.1136/heartjnl-2016-310894 PMID: 28432156
- Nishimura RA, Vahanian A, Eleid MF, Mack MJ. Mitral valve disease—Current management and future challenges. Lancet. Elsevier; 2016; 387: 1324–1334. https://doi.org/10.1016/S0140-6736(16)00558-4
- 4. lung B, Vahanian A. Epidemiology of valvular heart disease in the adult. Nat Rev Cardiol. Nature; 2011; 8: 162–172. https://doi.org/10.1038/nrcardio.2010.202 PMID: 21263455

- Rahimi K, Emdin CA, MacMahon S. The epidemiology of blood pressure and its worldwide management. Circ Res. 2015; 116: 925–36. https://doi.org/10.1161/CIRCRESAHA.116.304723 PMID: 25767281
- Singh JP, Evans JC, Levy D, Larson MG, Freed LA, Fuller DL, et al. Prevalence and clinical determinants of mitral, tricuspid, and aortic regurgitation (the Framingham Heart Study). Am J Cardiol. 1999; 83: 897–902. https://doi.org/10.1016/S0002-9149(98)01064-9 PMID: 10190406
- Herrett E, Gallagher AM, Bhaskaran K, Forbes H, Mathur R, van Staa T S L. Data Resource Profile: Clinical Practice Research Datalink (CPRD). Int J Epidemiol. 2015; 44: 827–36. https://doi.org/10.1093/ ije/dyv098 PMID: 26050254
- Walley T, Mantgani A. The UK General Practice Research Database. Lancet. 1997; 350: 1097–9. https://doi.org/10.1016/S0140-6736(97)04248-7 PMID: 10213569
- Lawrenson R, Todd JC, Leydon GM, Williams TJ, Farmer RD. Validation of the diagnosis of venous thromboembolism in general practice database studies. Br J Clin Pharmacol. Wiley-Blackwell; 2000; 49: 591–6. https://doi.org/10.1046/j.1365-2125.2000.00199.x PMID: 10848723
- Emdin CA, Anderson SG, Callender T, Conrad N, Salimi-Khorshidi G, Mohseni H, et al. Usual blood pressure, peripheral arterial disease, and vascular risk: cohort study of 4.2 million adults. BMJ. 2015; 351: h4865. https://doi.org/10.1136/bmj.h4865 PMID: 26419648
- Emdin CA, Anderson SG, Woodward M, Rahimi K. Usual Blood Pressure and Risk of New-Onset Diabetes: Evidence From 4.1 Million Adults and a Meta-Analysis of Prospective Studies. J Am Coll Cardiol. 2015; 66: 1552–62. https://doi.org/10.1016/j.jacc.2015.07.059 PMID: 26429079
- d'Arcy JL, Coffey S, Loudon MA, Kennedy A, Pearson-Stuttard J, Birks J, et al. Large-scale community echocardiographic screening reveals a major burden of undiagnosed valvular heart disease in older people: the OxVALVE Population Cohort Study. Eur Heart J. 2016; 37: 3515–3522. https://doi.org/10. 1093/eurheartj/ehw229 PMID: 27354049
- Prospective Studies Collaboration. Age-specific relevance of usual blood pressure to vascular mortality: a meta-analysis of individual data for one million adults in 61 prospective studies. Lancet. 2002; 360: 1903–1913. PMID: 12493255
- Rapsomaniki E, Timmis A, George J, Pujades-Rodriguez M, Shah AD, Denaxas S, et al. Blood pressure and incidence of twelve cardiovascular diseases: lifetime risks, healthy life-years lost, and age-specific associations in 1.25 million people. Lancet. 2014; 383: 1899–911. <u>https://doi.org/10.1016/S0140-6736(14)60685-1</u> PMID: 24881994
- Domanski M, Mitchell G, Pfeffer M, Neaton JD, Norman J, Svendsen K, et al. Pulse pressure and cardiovascular disease-related mortality: follow-up study of the Multiple Risk Factor Intervention Trial (MRFIT). JAMA. 287: 2677–83. PMID: 12020303
- Hutcheon JA, Chiolero A, Hanley JA. Random measurement error and regression dilution bias. BMJ. 2010; 340: c2289. https://doi.org/10.1136/bmj.c2289 PMID: 20573762
- Emdin CA, Anderson SG, Salimi-Khorshidi G, Woodward M, MacMahon S, Dwyer T, et al. Usual blood pressure, atrial fibrillation and vascular risk: evidence from 4.3 million adults. Int J Epidemiol. 2016; https://doi.org/10.1093/ije/dyw053 PMID: 27143136
- Easton DF, Peto J, Babiker AG. Floating absolute risk: an alternative to relative risk in survival and case-control analysis avoiding an arbitrary reference group. Stat Med. 1991; 10: 1025–35. PMID: 1652152
- 19. Honaker J, King G, Blackwell M. Amelia II: A Program for Missing Data. J Stat Softw. 2011; 45: 1–47. https://doi.org/10.18637/jss.v045.i07
- 20. Rubin D. Multiple imputation for non-response in surveys. New York: Wiley; 1987.
- 21. Woodward M. Epidemiology: study design and data analysis. 3rd Ed. Boca Raton: CRC Press; 2014.
- Vandenbroucke JP, von Elm E, Altman DG, Gøtzsche PC, Mulrow CD, Pocock SJ, et al. Strengthening the Reporting of Observational Studies in Epidemiology (STROBE): explanation and elaboration. PLoS Med; 2007; 4: e297. https://doi.org/10.1371/journal.pmed.0040297 PMID: 17941715
- 23. Emdin CA, Rothwell PM, Salimi-Khorshidi G, Kiran A, Conrad N, Callender T, et al. Blood Pressure and Risk of Vascular Dementia. Stroke.; 2016; 47: 1429–1435. https://doi.org/10.1161/STROKEAHA.116. 012658 PMID: 27165956
- Loardi C, Alamanni F, Trezzi M, Kassem S, Cavallotti L, Tremoli E, et al. Biology of mitral valve prolapse: the harvest is big, but the workers are few. Int J Cardiol.; 2011; 151: 129–35. <u>https://doi.org/10.1016/j.ijcard.2010.11.004</u> PMID: 21168228
- 25. Abramowitz Y, Jilaihawi H, Chakravarty T, Mack MJ, Makkar RR. Mitral Annulus Calcification. J Am Coll Cardiol.; 2015; 66: 1934–1941. https://doi.org/10.1016/j.jacc.2015.08.872 PMID: 26493666

- 26. Tobin MD, Sheehan NA, Scurrah KJ, Burton PR. Adjusting for treatment effects in studies of quantitative traits: antihypertensive therapy and systolic blood pressure. Stat Med. 2005; 24: 2911–2935. https://doi.org/10.1002/sim.2165 PMID: 16152135
- Clarke R, Shipley M, Lewington S, Youngman L, Collins R, Marmot M, et al. Underestimation of risk associations due to regression dilution in long-term follow-up of prospective studies. Am J Epidemiol. 1999; 150: 341–53. https://doi.org/10.1093/oxfordjournals.aje.a010013 PMID: 10453810